Author's response to reviews

Title: Iatrogenic insertion of impression mould into middle ear and mastoid and its retrieval after 9 years: A case report

Authors:

Mohammad Sohail Awan (sohail.awan@aku.edu)
Moghira Iqbal (moghiras@hotmail.com)
Zakariya Imam Sardar (zakariya.imam@gmail.com)

Version: 4 Date: 2 February 2007

Author’s response to reviews:

Title changed; A Case Report to A case report.
Author names corrected.
author qualifications removed.
All the 3 authors are from one place, appropriate change made.
reprint requests changed to corresponding author.
Abstract added.
keyword section removed from manuscript.
background section following abstract section.
reference citation in text format changed.
full stop after reference in text.
acknowledgement section moved before references.
Figure legend added.
reference of figure in text on page 5.